Author's response to reviews

Title: What Makes Public Health Studies Ethical? Dissolving the boundary between research and practice.

Authors:

Donald J Willison (don.willison@utoronto.ca)
Nancy Ondrusek (nancy.ondrusek@oahpp.ca)
Angus Dawson (a.j.dawson@bham.ac.uk)
Claudia Emerson (claudia.emerson@srcglobal.org)
Lorraine E Ferris (Lorraine.Ferris@utoronto.ca)
Raphael Saginur (rsaginur@ottawahospital.on.ca)
Heather Sampson (heather.sampson@utoronto.ca)
Ross Upshur (ross.upshur@gmail.com)

Version: 5
Date: 16 July 2014

Author's response to reviews:

2015-07-16

Thank you for your e-mail of 2014-07-16. Yes, there should have been a reference on line 21. It should be the same as that which appeared on line 22. This also caused me to identify an error in the reference itself. The first author should have been Jean Lin OTTO. For some reason, the reference imported into Reference Manager entered the first author as Lin Otto JEAN (wrong surname). Both corrections have been made: (1) I have now referenced Otto on both lines; (2) I have corrected the surname of the first author to “Otto” from “Jean”.

Thank you for catching this detail.

Don Willison